

Goal-directed osteoporosis treatment: ASBMR/BHOF task force position statement 2024

Felicia Cosman^{1,*}, E. Michael Lewiecki², Richard Eastell³, Peter R. Ebeling⁴, Suzanne Jan De Beur⁵, Bente Langdahl^{6,7}, Yumie Rhee⁸, Ghada El-Hajj Fuleihan⁹, Douglas P. Kiel^{10,11}, John T. Schousboe^{12,13}, Joao Lindolfo Borges¹⁴, Angela M. Cheung^{15,16}, Adolfo Diez-Perez¹⁷, Peyman Hadji¹⁸, Sakae Tanaka¹⁹, Friederike Thomasius²⁰, Weibo Xia²¹, Steven R. Cummings²²

¹Department of Medicine, Columbia University College of Physicians and Surgeons, New York, NY 10032, United States

- ²New Mexico Clinical Research & Osteoporosis Center, Division of Metabolic Bone Diseases, Albuquerque, NM 87106, United States ³Division of Clinical Medicine, University of Sheffield, Sheffield S10 2RX, United Kingdom
- ⁴Department of Medicine, School of Clinical Sciences, Monash University, Clayton, VIC 3168, Australia
- ⁵Division of Endocrinology and Metabolism, University of Virginia, Charlottesville, VA 22908, United States
- ⁶Department of Endocrinology, Aarhus University Hospital, Aarhus N 8200, Denmark
- ⁷Department of Clinical Medicine, Aarhus University, Aarhus N 8200, Denmark

⁸Department of Internal Medicine, Severance Hospital, Endocrine Research Institute, Yonsei University College of Medicine, Seoul 03722, South Korea

⁹Calcium Metabolism and Osteoporosis Program, WHO Collaborating Center for Metabolic Bone Disorders, American University of Beirut, Beirut 1107, Lebanon

- ¹⁰Marcus Institute for Aging Research, Hebrew Senior Life, Boston, MA 02131, United States
- ¹¹Department of Medicine, Beth Israel Deaconess Medical Center, Harvard Medical School, Boston, MA 02215, United States
- ¹²Division of Research and Evaluation, HealthPartners Institute, Bloomington MN 55425, United States
- ¹³Division of Health Policy and Management, University of Minnesota, Minneapolis, MN 55425, United States
- ¹⁴Brazil Clinical Research Center, Brasilia 71625-175, Brazil
- ¹⁵Department of Medicine and Joint Department of Medical Imaging, University Health Network, Toronto, ON M5G 2C4, Canada
- ¹⁶Centre of Excellence in Skeletal Health Assessment, University of Toronto, Toronto, ON M5G 2C4, Canada
- ¹⁷Department of Medicine, Institute Hospital del Mar of Medical Investigation, Barcelona 08003, Spain
- ¹⁸Department of Obstetrics and Gynecology, Frankfurt Center of Bone Health and Phillipps-University of Marburg, Frankfurt 60313, Germany
- ¹⁹Department of Orthopaedic Surgery, Faculty of Medicine, The University of Tokyo, Hongo, Bunkyo-ku, Tokyo 113-0033, Japan
- ²⁰Department of Clinical Osteology, Frankfurt Center of Bone Health and Endocrinology, Frankfurt 60313, Germany

²¹Department of Endocrinology, Peking Union Medical College Hospital, Chinese Academy of Medical Sciences & Peking Union Medical College, Beijing 100730, China

²²San Francisco Coordinating Center, CPMC Research Institute, Department of Epidemiology and Biostatistics, University of California, San Francisco, CA 94158, United States

*Corresponding author: Felicia Cosman, Professor of Medicine, Columbia University College of Physicians and Surgeons, 630 West 168th St, New York, NY 10032, United States (felcosman@gmail.com).

Abstract

The overarching goal of osteoporosis management is to prevent fractures. A goal-directed approach to long-term management of fracture risk helps ensure that the most appropriate initial treatment and treatment sequence is selected for individual patients. Goal-directed treatment decisions require assessment of clinical fracture history, vertebral fracture identification (using vertebral imaging as appropriate), measurement of bone mineral density (BMD), and consideration of other major clinical risk factors. Treatment targets should be tailored to each patient's individual risk profile and based on the specific indication for beginning treatment, including recency, site, number and severity of prior fractures, and BMD levels at the total hip, femoral neck, and lumbar spine. Instead of first-line bisphosphonate treatment for all patients, selection of initial treatment should focus on reducing fracture risk rapidly for patients at very high and imminent risk, such as in those with recent fractures. Initial treatment selection should also consider the probability that a BMD treatment target can be attained within a reasonable period of time and the differential magnitude of fracture risk reduction and BMD impact with osteoanabolic versus antiresorptive therapy. This position statement of the ASBMR/BHOF Task Force on Goal-Directed Osteoporosis Treatment provides an overall summary of the major clinical recommendations about treatment targets and strategies to achieve those targets based on the best evidence available, derived primarily from studies in older postmenopausal women of European ancestry.

Keywords: practice/policy-related issues, anabolics, therapeutics, antiresorptives, therapeutics, osteoporosis, diseases and disorders of/related to bone, DXA, analysis/quantitation of bone

Received: March 5, 2024. Revised: June 24, 2024. Accepted: July 4, 2024

© The Author(s) 2024. Published by Oxford University Press on behalf of the American Society for Bone and Mineral Research.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Lay Summary

Goal-directed treatment can help healthcare providers recommend the best treatments for individual patients to prevent fractures. The goaldirected strategy considers the site, number, and recency of prior fractures. This may require imaging for spine fractures, which may not have caused pain. Treatment decisions also require bone mineral density (BMD) measurement and consideration of other major risk factors. In contrast to the standard approach, same first treatment for all, treatment selection is tailored to an individual's risk. In patients with recent fractures of the spine, hip, or pelvis, fracture risk is very high and treatment should rapidly reduce that risk. For others, the target is a specific BMD level and should consider the likelihood that the treatment target can be attained within a reasonable period of time, which differs for osteoporosis medications. After initial therapy, BMD should be assessed to determine if the target has been achieved. If so, strategies should focus on maintaining BMD. If the target is not yet achieved, treatment should be intensified, or continued if it is already the most potent option. This position statement represents a consensus of expert recommendations about treatment targets and strategies to achieve those targets based on the best available evidence.

Introduction

Since our last report in 2017,¹ the emergence of new medical evidence and development of new therapeutic agents have prompted a need to revisit the concept of goal-directed treatment (treat-to-target) for osteoporosis. For these reasons, the American Society for Bone and Mineral Research (ASBMR) and Bone Health & Osteoporosis Foundation (BHOF, formerly National Osteoporosis Foundation) re-assembled an international task force of clinical scientists and expert physicians to update the previous report and develop a pragmatic approach to aid clinicians in managing patients with osteoporosis. Multiple drafts of this document were extensively vetted by all authors. Comments, suggestions and edits were carefully considered and text was accepted or modified until all authors were in agreement. Pharmaceutical companies provided no funding for the development of this manuscript and had no role in its concept, writing, or revisions. There was no external funding from any source. All coauthors approved the final version of the position statement.

The most recent guidelines from the American Association of Clinical Endocrinology, Endocrine Society, North American Menopause Society (now The Menopause Society), BHOF, European Society for Clinical, and Economic Aspects of Osteoporosis and Osteoarthritis/International Osteoporosis Foundation and National Osteoporosis Guideline Group²⁻⁷ have included a new very high fracture risk category. A subset of very high-risk patients is also at imminent risk of fracture for 2 yr following an incident fracture^{2,4,5,8-16} or after multiple prior fractures.^{2,17-20}

Since the last report, two new osteoanabolic agents, abaloparatide and romosozumab, have been approved for treatment of patients with osteoporosis in the USA and in many other countries. Head-to-head studies in high or very high-risk patients demonstrate that osteoanabolic agents are more effective in preventing osteoporotic fractures than bisphosphonates²¹⁻²⁵ and denosumab.²⁶ Additionally, the Foundation for the National Institutes of Health/Study to Advance bone mineral density (BMD) as a Regulatory Endpoint collaboration (FNIH/SABRE) showed a strong relationship between BMD increases with treatment compared with placebo, and the magnitude of anti-fracture efficacy in a meta-regression of over 50 clinical trials.²⁷⁻²⁹ In clinical trials, the increase in total hip (TH) and femoral neck (FN) BMD in treated patients compared with placebo explained a somewhat greater proportion of the anti-fracture effect than change in lumbar spine (LS) BMD.²⁸

Data from trials with antiresorptive and osteoanabolic agents show an inverse relationship between the BMD level achieved in patients on treatment and the subsequent risk of fracture.³⁰⁻³³ New evidence shows that TH BMD appears to be the most useful treatment target because it consistently

predicts the risk of both vertebral and nonvertebral fractures, whereas the LS BMD level after treatment predicts the risk of vertebral fracture but does not predict the risk of nonvertebral fractures as consistently.³³ In addition, new evidence provides guidance for identifying a BMD target.^{32,33} Other studies have confirmed the benefit of treating with an osteoanabolic medication before an antiresorptive drug, compared with the reverse order, to maximize gains in BMD, particularly in the hip³⁴⁻³⁸ and the probability of achieving different BMD treatment targets beginning with osteoanabolic rather than antiresorptive agents at varying starting BMD levels.³⁹⁻⁴¹

It is common clinical practice and a requirement of many health insurers to prescribe an oral bisphosphonate as initial treatment for all patients with osteoporosis, unless a contraindication is present. However, this "step therapy" approach does not provide the most effective treatment for all high and very high-risk patients. In contrast, goal-directed treatment individualizes treatment decisions based on an individual's fracture risk, taking into account fracture history, BMD, and other risk factors, aiming at prespecified individualized treatment targets. Based on these considerations, some patients will benefit most from osteoanabolic therapy as initial treatment.

Goal-directed therapy is a strategy for the long-term management of patients receiving treatment for osteoporosis. Achieving treatment targets might require intensification of therapy if a fracture occurs or the patient remains far from a BMD target despite osteoporosis treatment. This intensification could include replacing a bisphosphonate with denosumab, replacing a bisphosphonate or perhaps denosumab with an osteoanabolic agent, or adding an osteoanabolic agent to ongoing treatment with a bisphosphonate or denosumab. It might also include a repeat course of osteoanabolic medication. It must be acknowledged, however, that the BMD effects of switching from antiresorptive to osteoanabolic agents are not as robust as those seen when initiating treatment with an osteoanabolic agent (especially when switching from denosumab).³⁸ In addition, the evidence supporting the safety and efficacy of repeat courses of osteoanabolic agents is extremely limited. Extended use beyond 24 mo and repeated use of branded teriparatide is approved in the USA when a patient remains at or has returned to having a high risk for fracture.

This report represents the consensus of the ASBMR/BHOF Task Force on Goal-Directed Osteoporosis Treatment, based on interpretation of the best evidence available. This position statement is not a clinical guideline. It begins with a summary overview of the main principles, followed by a description of the evidence underlying these principles. It is focused on postmenopausal White women for whom the vast majority of evidence is available. For physicians who are not confident prescribing goal-directed therapy or managing very-high risk patients, referral to an osteoporosis specialist should be considered.

Summary of major principles

Treatment targets

Treatment targets should be individualized, based in large part on the specific indication for beginning treatment, including recency of fracture, number and site of prior fractures, severity of vertebral fracture(s), BMD at the TH, FN, and/or LS, age, and other strong risk factors for fracture.

- Patients at imminent risk of fracture: Patients with a recent fracture are at very high risk of more fractures over the next 2 yr; this risk is largely independent of baseline T-score. The treatment goal for these patients is to rapidly and maximally reduce fracture risk. Greater BMD increases are associated with greater reduction in fracture risk. Some patients who have had multiple prior fractures (even if not within the last 2 yr) are also at imminent risk. Many of these patients, the treatment goal is rapid and maximal reduction in fracture risk, but more sustained treatment is likely required than for a single recent fracture.
- For patients who are not at imminent risk of fracture, baseline T-scores, fracture history, and other major risk factors guide the selection of treatment targets.
- Patients with baseline T-score ≤ -2.5 : For patients with baseline T-scores < -2.5 at the TH, FN, and/or LS, the treatment target is a T-score level at least > -2.5 at the respective skeletal sites.
 - In patients on osteoporosis treatment, TH T-score best reflects subsequent fracture risk at both vertebral and nonvertebral sites. In patients on osteoporosis treatment, T-score at the FN also reflects subsequent risk of fracture at both vertebral and nonvertebral sites. Since reproducibility is better for the TH than the FN, it is the preferred region of interest for a treatment target. For patients with an isolated T-score < -2.5 at the FN, the FN T-score can be the treatment target. Improving T-score to levels > -2.5 is associated with lower risk for nonvertebral and vertebral fractures. A T-score > -2.5 should be the minimum treatment target. In some patients, achieving a higher T-score target might be warranted, based on other risk factors.
 - In patients on osteoporosis treatment, LS T-score reflects subsequent risk of vertebral fracture, but the association with nonvertebral fracture is less consistent. Improving LS T-score to levels > -2.5 is associated with lower vertebral fracture risk; a T-score > -2.5 should be the minimum treatment target. In some patients, achieving a higher target T-score might be warranted based on other risk factors.
 - For countries with different T-score intervention thresholds, T-score targets should be adjusted accordingly.
- Patients with baseline T-score > -2.5: For patients with baseline T-scores > -2.5 who are recommended for treatment because of high fracture risk (such as prior fracture or high-risk medical conditions), increasing BMD remains associated with reduced fracture risk. Greater BMD increases with larger T-score improvements are

associated with greater fracture-risk reduction, but a T-score target is not easily defined in these patients.

Selecting treatment to achieve treatment targets

- *Initial treatment:* Selection of initial treatment should consider the probability that a treatment target can be attained over a reasonable period of time, with greater urgency for patients at imminent fracture risk (recent fracture or some multiple prior fractures). Data to guide these decisions include the likelihood that a treatment can provide at least a 50% probability of attaining the T-score target over 3 yr, depending on the initial BMD. For some patients, it might be appropriate to select treatment to achieve a higher T-score target, reach the treatment target faster, or provide a higher probability of achieving the treatment target.
- Sequence of treatment: Treatment sequence is important for attaining T-score targets particularly in patients with baseline T-scores far below the treatment target. Selection of initial therapy should acknowledge the differential impact on BMD for osteoanabolic-first versus antiresorptive-first treatment sequences. Greater and faster BMD increases at the hip and spine are seen with osteoanabolic-first followed by antiresorptive treatment strategies, compared with the reverse order.

Determining if treatment targets have been achieved

To determine whether a treatment target has been achieved requires repeat BMD testing and assessment for new fractures, including vertebral imaging. Having a baseline vertebral image before starting treatment allows confirmation that an incident vertebral fracture has occurred on follow-up vertebral imaging. If a patient experiences one or more new fractures, it indicates that the most important treatment target has not been met, regardless of the T-scores achieved. Monitoring for achievement of a treatment target is distinct from monitoring for treatment responses. To achieve a treatment target, a treatment response is necessary. Treatment response, which can be assessed by measurement of biochemical turnover markers, depends on patient adherence and type and duration of prior osteoporosis medication.

- When a treatment target has not been achieved or is unlikely to be achieved, consider changing to more potent therapy (or continuing the highest potency treatment sequence).
- When a treatment target has been reached, maintenance of BMD is the focus. This may involve continuing treatment, changing treatment, pausing treatment, or administering intermittent bisphosphonates.

Review of the evidence to support major principles and actionable steps in patient management

To establish a treatment target and select initial therapy requires assessment of fracture risk, including vertebral imaging when appropriate, and BMD measurement when available. Indications for BMD testing and vertebral imaging are available in the BHOF Clinician's Guide⁶ and the International Society for Clinical Densitometry Position Statements.⁴² Treatment targets should reflect the underlying clinical scenarios; specific BMD



Figure 1. One-year risk of recurrent fracture in women > 65 years of age, based on site of initial fracture, from Medicare database of 377,561 women with first fracture. Adapted from reference 8.

treatment targets and choice of therapeutic agents should be individualized.

Treatment targets and selection of treatment for patients at imminent risk

Recent fracture and imminent risk of subsequent fracture

Multiple recent studies in different regions of the world confirm the importance of recent fracture as a predictor of subsequent fracture.⁸⁻¹⁶ Overall, these large database studies show that the risk of fracture is increased dramatically for up to 2 yr after the occurrence of the first fracture, with the annual risk remaining high but declining after that time. Risk increases very rapidly after an initial fracture, with the highest risk seen within the first few months after the fracture.¹⁰ One of the most comprehensive studies utilized a Medicare database to evaluate 377,500 women age 65 yr and older who had a first clinical fracture.⁸ Women with fractures of the digits, skull, patella, sternum, scapula, and ribs were excluded. Fracture risk was assessed for up to 5 subsequent yr. Average risk of subsequent fracture was 10% in the very next year and 18% in the 2 yr following the first fracture. The site of the first fracture was associated with varying risks of subsequent fracture (Figure 1).⁸ The magnified short-term risk was particularly high after vertebral and pelvic fractures and lowest for ankle fractures, though even ankle fractures were still associated with a subsequent 1-yr risk of just below 5% and 2-yr risk of almost 10% (which is still considered imminent risk).43 Therefore, in patients at imminent risk, especially those with recent fractures of the spine, hip, and pelvis, rapid and maximum fracture risk reduction is the first and most important treatment target.

Multiple prior fractures and imminent risk of subsequent fracture

Patients with multiple prior fractures may also be at imminent risk for more fractures.^{2,17-20} Fracture site, severity, and time from last fracture occurrence remain important determinants of subsequent risk. In many patients with two or more fractures, rapid and maximum fracture risk reduction is the most important treatment goal. This is particularly pertinent for women with fracture sites that include spine, hip, or pelvis and for patients whose fractures were not remote (more than 10 yr earlier). Additionally, in many patients with multiple fractures, very-high risk persists, especially if there are no modifiable risk factors. Treatment targets might include attaining higher T-score targets than specified below. Furthermore, these patients might require a more prolonged and intensive course of therapy even after achieving T-score targets.

Selecting initial treatment for patients at imminent risk

Since osteoanabolic agents reduce fracture risk faster and to a greater extent than antiresorptive agents,²¹⁻²⁶ osteoanabolic therapy may be better for patients at imminent fracture risk, especially after recent major fractures such as spine, hip, and pelvis (Figure 1).⁸ Initiating treatment with osteoanabolic agents followed by antiresorptive agents also increases BMD more than the reverse treatment sequence. Therefore, BMD treatment targets are more likely to be achieved rapidly with an osteoanabolic-first treatment sequence. For patients with other recent fractures, osteoanabolic agents, bisphosphonates, or denosumab may be appropriate, depending site of fracture and BMD. A consensus regarding which recent nonhip, non-spine, non-pelvis fractures should prompt use of osteoanabolic therapy versus bisphosphonates or denosumab is needed.

Treatment targets and selection of treatment for patients with T-scores ≤ -2.5

Why is BMD a good target for osteoporosis treatment and what is the best skeletal site?

In a large meta-regression of pharmaceutical clinical trials, 2-yr mean BMD differences between active and placebotreated patients in the TH, FN, and LS were associated with fracture risk reduction;²⁷⁻²⁹ larger mean BMD increases predicted greater reductions in fracture risk. Mean TH and FN BMD increments explained a somewhat larger proportion of fracture risk reduction than mean LS BMD increments (Table 1).²⁸ LS BMD is more likely affected by degenerative artifact and aortic calcification. Of the two hip regions of interest, the TH has better reproducibility than the FN.²⁸

Mean TH BMD level after treatment is predictive of subsequent risk of both nonvertebral and vertebral fractures.³⁰⁻³³ The relationship between mean LS BMD level after treatment has not been tested in as many clinical trials, since several studies did not assess LS BMD in all individuals at all time

Table 1. Proportion of treatment-related fracture reduction effect explained by BMD increment at the TH, FN, and LS (95% CI).

	Total Hip	Femoral Neck	Lumbar Spine
Vertebral fracture Nonvertebral fracture	59% (50-69) 63% (38-88)	61% (51-72) 67% (40-95)	31% (19-44) 52% (23-82)
Hip fracture	48% (21-76)	44% (12-77)	42% (9-75)

Adapted from reference 28.

points.^{30,31} In trials where LS BMD has been measured in the full population, mean LS T-score reflects subsequent risk of vertebral fracture consistently; however, the association with nonvertebral fracture is not universal. For example, in a post hoc analysis from the ARCH study (Active-controlled fracture study in postmenopausal women with osteoporosis at high risk), mean LS BMD attained after 1 yr of treatment with alendronate or romosozumab was associated with subsequent risk of vertebral fracture but not nonvertebral fracture (Figure 2).³³

For all of these reasons, the single best skeletal site for a T-score target is the TH. This might be important in some clinical situations, for example where the TH T-score is already > -2.5, but LS T-score is below. Medication choice might differ in this type of patient, rather than the reverse situation where LS T-score is > -2.5, but TH T-score is lower (see below). In patients who have only a FN T-score < -2.5 or a LS T-score < -2.5, a T-score > -2.5 at the respective skeletal site should be the treatment target.

What is the rationale for choosing the minimum target of > -2.5?

In some countries, a T-score < -2.5 represents an indication for pharmacologic treatment regardless of other risk factors.⁶ However, since fracture risk is dependent on other factors, notably age and prevalent fracture,^{44,45} a T-score > -2.5should be considered the *minimal target*. For countries with different or no T-score intervention thresholds, T-score targets should be adjusted accordingly.⁴⁶

Patients with other important risk factors might warrant higher T-score targets. This is demonstrated very clearly in analyses from the denosumab FREEDOM trial (Fracture Reduction Evaluation of Denosumab in Osteoporosis Every 6 Mo) and its extension.³² The relationship between attained T-score on denosumab treatment and subsequent fracture risk is maintained in individuals with and without prior fracture, but the risk at all BMD levels is much higher in those with a fracture history (Figure 3).³² These data support setting higher T-score targets in patients with a history of fracture. Higher T-score targets might also be suggested for patients with advanced age, recent falls history, and poor physical function.⁴⁷⁻⁴⁹ In these patients, T-score targets of -2.0 or even -1.5 might be appropriate. In the ARCH trial, where all patients had fractures at entry, progressive lowering of subsequent fracture risk was observed as 1 yr TH T-scores approached $-1.5.^{33}$

A higher T-score target might also be considered in patients for whom a pause in medication is being considered. Anyone who stops osteoporosis medication will lose bone eventually, though the rate of loss is dependent on what medication is being discontinued. Bone loss rates after stopping bisphosphonates are much slower than bone loss rates after stopping all other agents. Therefore, medication pauses should only be considered in patients whose last treatment is a bisphosphonate. Aiming for a higher T-score in those patients being considered for a medication pause might permit maintenance of Tscore above -2.5 even after the expected bone loss occurs (see section below on BMD maintenance for further discussion).

Selecting initial treatment to achieve target BMD

The choice of therapeutic agent is dependent on which skeletal site has the lowest T-score and exactly what the starting BMD is at that site. When selecting initial treatment, the clinician should consider the probability that BMD targets can be attained with a specific treatment over a reasonable period of time (see "Probability of Attaining BMD Treatment Targets" below). Selection of initial therapy should also acknowledge the differential impact on BMD for osteoanabolic-first versus antiresorptive-first treatment sequences and the differential effects of medications on hip and spine BMD.

Goal-directed therapy choices should also consider other important clinical factors:

- In patients with TH T-score -2.5 to -2.8 (inclusive) and LS T-score -2.5 to -3.0 (inclusive): bisphosphonates, denosumab, and osteoanabolic agents are all likely to enable the achievement of T-scores > -2.5 at both TH and LS within 3 yr. Treatment choices should consider a patient's fracture history, including the skeletal site and timing of prior fracture. For patients with no prior fracture or a fracture other than spine, hip, or pelvis, bisphosphonates and denosumab are both highly likely to enable achievement of treatment targets within 3 yr and are the most appropriate choice. In contrast, in patients who have had a major fracture such as the spine, hip, or pelvis, osteoanabolic agents should be considered. This is most important if those fractures occurred somewhat recently (eg, more than 2 yr before, but within the last decade). In those patients, choosing osteoanabolic agents as initial therapy might enable patients to achieve treatment targets faster, to achieve higher T-score targets, or to have a higher probability of achieving the treatment target when beginning treatment.³⁹⁻⁴¹ Individual patient factors and local/regional guidance should be considered when deciding treatment based on site of prior fracture or T-score intervention threshold.⁴⁶
- In patients with very low BMD (TH T-scores < -2.8or LS T-scores < -3.0): osteoanabolic agents might be required to achieve BMD treatment targets (see next section on probability of attaining BMD targets). Osteoanabolic agents and osteoanabolic-first treatment sequences produce larger BMD gains in both TH and LS compared with antiresorptive agents alone. Therefore, osteoanabolic agents should be considered as initial therapy for these patients when possible. In patients who start with a TH Tscore < -3.5, it might be impossible to achieve a treatment target with any current medication in 3 yr; however, treatment decisions should aim to improve TH T-score to levels as close to the target as possible. Bisphosphonates or denosumab might be the first choice in some patients with no other risk factors, especially in countries where T-score intervention thresholds are lower.

Probability of attaining BMD treatment targets

Table 2 provides estimates of the lowest starting TH or LS Tscore that would allow at least a 50% probability of attaining



Figure 2. Relationship between T-score attained after one year of treatment with either alendronate or romosozumab and subsequent risk of nonvertebral (left) or vertebral fracture (right) for the total hip (A and B), femoral neck (C and D) and lumbar spine (E and F).³³

a T-score > -2.5 at that site, based on data available for different therapeutic agents over ~ 3 yr of treatment. These data can be helpful in making initial treatment decisions.

Probabilities with romosozumab followed by alendronate or denosumab versus alendronate only

In a post hoc analysis of the ARCH and FRAME (Fracture study in postmenopausal women with osteoporosis) studies, the probability of attaining BMD targets over a total of 3 yr

of therapy with alendronate alone, 1 yr of romosozumab followed by alendronate, or 1 yr of romosozumab followed by denosumab was determined (Figure 4; Table 2).³⁹ In patients with TH T-score of -2.7, the probability of achieving a TH T-score > -2.5 with alendronate alone over 3 yr was > 50%, whereas this probability was only 9% for those who started with TH T-score of -3.0. For romosozumab followed by alendronate, the corresponding probabilities were 73% and 38%, respectively. For romosozumab followed by denosumab, these probabilities were 90% and 61%, respectively.³⁹



Figure 3. Relationship between T-score attained on denosumab and subsequent risk of nonvertebral fracture in full population (A), by age < 75 vs. > 75 yr (B), and by prior history of fracture (C).³²

Table 2. Lowest baseline T-score that permits > 50% of women to achieve a T-score > -2.5 in approximately 3 yr.

	Total Hip	Lumbar Spine
Alendronate	-2.7	-3.0
Denosumab	-2.8	-3.1
Romosozumab/Alendronate	-2.9	-3.5
Abaloparatide/Alendronate	-2.9	-3.5
Romosozumab/Denosumab	-3.1	-3.7

Adapted from references 39-41.

For women who began with LS T-scores of -3.0, the probability of achieving a LS T-score above -2.5 with alendronate alone was 55%, while it was only 11% with a starting LS T-score of -3.5. With romosozumab followed by alendronate, these probabilities were 81% and 46% respectively for starting T-scores of -3.0 and -3.5, respectively. With romosozumab followed by denosumab, the probabilities were 93% and 67%, respectively.

Probabilities with abaloparatide for 18 mo followed by alendronate for 2 yr

In a post hoc analysis of the ACTIVE study (Abaloparatide Comparator Trial in Vertebral Endpoints) and its extension,

the probability of attaining BMD targets with abaloparatide for 18 mo followed by alendronate for 2 yr was determined.⁴⁰ Women with baseline TH T-scores as low as -2.9 and baseline LS T-scores as low as -3.5 had a > 50% probability of attaining a T-score above -2.5 at each respective site with abaloparatide followed by alendronate (Table 2).

Probabilities with denosumab over 3 yr

In a post hoc analysis of the FREEDOM study, the probability of attaining BMD targets with 3 yr of denosumab treatment was determined.⁴¹ There was a > 50% probability that women with starting TH T-scores > -2.8 and LS T-scores > -3.1 could achieve target T-scores above -2.5 with 3 yr of denosumab treatment (Table 2).

Importance of treatment sequence

BMD increases with osteoanabolic agents are lower, particularly in the TH, when patients transition from a bisphosphonate or denosumab. Upon transition from a bisphosphonate to teriparatide, TH BMD declines consistently for at least 12 mo and hip strength is not improved for at least 1 yr.^{34,35} Data are not currently available to assess the BMD effects of a treatment sequence beginning with an antiresorptive and switching to abaloparatide. When patients on denosumab are switched



Figure 4. Probability of attaining T-score above -2.5 for TH (A) and LS (B) over 3 yr during treatment with alendronate, romosozumab followed by alendronate, or romosozumab followed by denosumab, based on starting T-score.³⁹

to teriparatide, TH BMD declines rapidly and remains below the on-denosumab baseline for a full 2 yr.³⁶ Upon transition from a bisphosphonate to romosozumab, TH BMD and hip strength improve³⁵ although not as markedly as seen with romosozumab in treatment naïve individuals. After 1 yr of denosumab treatment, upon transition to romosozumab, TH BMD remains stable;⁵⁰ however, the BMD effect with romosozumab might differ after a longer preceding denosumab treatment duration. LS BMD increments are also lower with osteoanabolic agents upon transition from antiresorptive agents; however, the difference is not as marked as it is in the TH.^{34-38,50}

Choice of treatment after osteoanabolic

In patients initially treated with osteoanabolic therapy for 1-2 yr (as appropriate for the agent used), the subsequent choice of antiresorptive agent should take into account the repeat BMD level and other clinical risk factors. Patients who are still far from BMD targets might benefit from denosumab to attain BMD targets, since denosumab increases BMD more than bisphosphonates after osteoanabolic treatment.³⁸ In patients who are close to BMD targets after osteoanabolic treatment, intravenous or oral bisphosphonates can be considered as follow-up therapy. In some countries, it might be possible to consider another course of osteoanabolic therapy some years later for those who still have T-scores below -2.5. If another course of osteoanabolic therapy is foreseen, bisphosphonates may be the better intermediate treatment choice. This would avoid the overshoot bone remodeling seen after denosumab discontinuation, which might mitigate the effect of the second course of osteoanabolic treatment.³⁸

Treatment targets and selection of treatment for patients with T-scores > -2.5

More than half of all patients who have adulthood fractures have BMD levels above osteoporosis range.⁵¹⁻⁵³ For patients with baseline T-scores > -2.5 who have had a single prior fracture that occurred more than 2 vr earlier, subsequent risk might differ substantially by skeletal site and time since fracture. Prior vertebral, hip, and pelvic fractures are associated with higher and more persistent risk than other fractures. For higher-risk patients who require treatment, increasing BMD is associated with reduced fracture risk.⁵⁴ There is a paucity of evidence to guide the actual BMD level to target in these patients. Percentage increase in BMD is a function of baseline BMD, with lower percentage increases expected in those with higher baseline BMD.54 Another level of uncertainty is how percentage changes in BMD relate to T-score unit changes. These relationships are specific for different populations at different starting BMD levels. For example, in the ARCH study, where all women began with T-scores < -2.0, mean BMD percentage increments with 1 yr of alendronate or romosozumab treatment were compared with mean Tscore changes (Table 3).³³ With 1 yr of alendronate, BMD increments of 2.9% in TH and 5.1% in LS corresponded to T-score unit changes of 0.2 and 0.3, respectively. With romosozumab, BMD increases of 6% in TH and 14% in LS BMD were associated with T-score unit increases of 0.3 and 0.9. BMD gain and T-scores continue to increase over at least 1-2 subsequent yr. TH T-score increments of at least 0.2 units and LS increments of at least 0.5 units are achievable over 3 yr with most therapies (bisphosphonates, denosumab, and osteoanabolic agents) and would be expected to reduce risk of both vertebral and nonvertebral fracture.⁵⁵⁻⁶² Therefore,

Table 3. Relationship between % BMD increase and T-score change in the TH and LS at 1 yr after treatment with romosozumab or alendronate.

	Romosozumab (n=1739)		Alendronate (n = 1726)	
	Mean % change BMD	Mean change T-score	Mean % change BMD	Mean change T-score
Total hip	6.3%	0.31	2.9%	0.15
Lumbar spine	13.9%	0.90	5.1%	0.34

Adapted from reference 33.



- · For imminent risk patients, maximal rapid reduction in fracture risk
- For patients with T-score ≤ -2.5, minimal target is to increase T-score to > -2.5, higher for patients with fracture history, or other major risk factors
- For patients with T-score > -2.5, increase TH T-score by 0.2 (3%) and LS by 0.5 (6%)



Figure 5. Goal directed therapy algorithm. Treatment targets and initial treatment selection, guided by fracture history (site, number and recency) and BMD. ¹Risk likely differs based on fractures that occurred a few years earlier vs. very remote events (eg 15 yr earlier). ²Many, but not all, patients with multiple fractures are at imminent risk, based on fracture sites and time from fracture occurrence. ³In some countries, T-score intervention thresholds are lower; T-score targets should be adjusted accordingly. ⁴There is no consensus about which recent fracture sites should be recommended for osteoanabolic treatment vs. BP or DMAB or what T-score should prompt use of osteoanabolic treatment in patients with recent "other fractures." ⁵BP or DMAB might be first choice in some patients with no other risk factors, especially in countries where T-score intervention thresholds are lower. ⁶Choosing osteoanabolic agents provides a higher probability of achieving treatment targets, achieving treatment targets faster, and achieving higher T-score targets. Abbreviations: TH, total hip; FN, femoral neck; LS, lumbar spine; BP, bisphosphonates; DMAB, denosumab.

these T-score changes can be considered treatment targets in these patients.

For patients with no prior fracture, T-score above -2.5, but high fracture probability according to a fracture risk algorithm, initiation of treatment with an antiresorptive agent may be appropriate, with a goal of increasing TH T-score by 0.2 (3%) and LS T-score by 0.5 (6%).

An algorithm, which presents treatment targets and principles guiding selection of initial treatments for individual patients, is provided in Figure 5.

Determining if treatment targets have been achieved

Determining whether patients have achieved treatment targets is distinct from monitoring for treatment responses; patients might have had a good treatment response yet still be at high risk of fracture if treatment targets have not been achieved.

When treatment targets have not been met

Patients with clinical fractures or confirmed vertebral fractures while on medication have not achieved treatment targets and are at imminent risk of more fractures. While a fracture on treatment does not necessarily represent treatment failure, it does suggest that fracture risk is higher than previously recognized. After an evaluation for secondary causes of osteoporosis, interventions to reduce falls risk and improve treatment adherence should be implemented and patients should be started or continued on the most potent medication (or medication sequence) for at least the next 2 yr. This strategy could also consider switching to or adding an osteoanabolic agent to ongoing antiresorptive medication. In women who have already been treated with osteoanabolic medication and remain at very high risk, another course of osteoanabolic medication (either using the same or different medication) could be considered. The optimal timing of a second course of treatment and how to maximize BMD gain after antiresorptive therapy is unknown, although repeated use of branded teriparatide is approved in some countries, if a patient remains at or has returned to a high risk for fracture. This approach might also be appropriate for women whose BMD levels remain below the target or those who lose BMD despite therapy. Continuing an antiresorptive medication, such as denosumab, is also a potential strategy, especially when osteoanabolic options are limited.⁶³

When treatment targets have been achieved

For patients who have achieved treatment targets, benefits are expected to wane if medication is discontinued. BMD declines rapidly after discontinuation of most agents and important clinical consequences might occur, such as multiple vertebral fractures after denosumab discontinuation.^{64,65} Upon cessation of denosumab, after short-term use (< 3 yr), transition to bisphosphonates appears to be effective at mitigating bone loss and risk of multiple vertebral fractures, but the most effective regimen after long-term use remains uncertain.^{38,64,65} In contrast to denosumab discontinuation, with cessation of bisphosphonate treatment, residual effects can persist for several years after discontinuation,^{66,67} particularly for zoledronic acid.⁶⁷ Therefore, when treatment targets have been met with a nonbisphosphonate, transitioning to a bisphosphonate may serve to maintain BMD gains. In one study of women with low bone mass, a single infusion of zoledronic acid was able to maintain BMD for up to 5 yr.⁶⁸ Administration of periodic zoledronic acid, with a dose interval beyond 1 yr, might therefore be useful for long-term maintenance of treatment targets achieved. Periodic administration of oral bisphosphonates might also permit maintenance of BMD targets. For patients who have achieved treatment targets with bisphosphonate therapy, a pause in treatment ("bisphosphonate holiday") may be considered. More information is needed to clarify the optimal dosing regimen to avoid rare but important side effects associated with bisphosphonate duration, such as atypical femur fractures and osteonecrosis of the jaw, while maximizing the likelihood of maintaining BMD.

Limitations

10

The evidence used here is based almost solely on women self-reporting as White, primarily 60 yr of age and older. It needs to be clarified how recommendations should differ in lower risk populations (younger patients; men; women of other ethnicities, race, or geographical location). Furthermore, specific BMD levels are associated with a wide variation in absolute fracture risk, depending on ethnicity and geography. Indications for treatment vary in different regions of the world and the treatment targets suggested here should be modified according to local considerations. The treatment targets might also differ in patients with secondary osteoporosis conditions such as glucocorticoid-induced osteoporosis or in patients with diabetes mellitus.

A consensus is needed to confirm which fractures constitute a risk high enough to warrant initial use of osteoanabolic therapy. This is true for both patients with recent fractures and fractures that occurred more than 2 yr earlier. Consensus is also needed on defining an acceptable level of fracture risk after treatment. Reducing fracture risk is the treatment goal; currently, BMD targets are the best way to determine if this goal has been achieved. Determining whether BMD treatment targets have been met requires good quality, reliable BMD data, and interpretation. With a very low starting Tscore, especially in the TH, treatment targets might not be attainable even with the most potent currently available therapy; however, the concept of choosing treatment sequences likely to attain BMD levels close to the target is still appropriate. The data to derive probabilities of attaining treatment targets are from post-hoc analyses, with two of the studies published only as abstracts at the current time. There is a paucity of evidence about treatment targets for patients who have fractures with starting T-scores above -2.5. Making treatment decisions based on at least a 50% probability of achieving the BMD target is intended as a general guide, not a rule.

The approach presented in this position statement does not take into account limitations imposed by health systems and insurers, cost, cost-effectiveness, or patient preferences. Osteoanabolic agents are relatively expensive (vs. antiresorptive agents) and often require a pre-approval process. Medication authorization, insurance coverage, medication cost, and out of pocket costs to consumers vary considerably around the world. These factors must be considered when treatment decisions are discussed with individual patients. There are few data addressing the optimal timing and effectiveness of repeat courses of osteoanabolic treatment after antiresorptive medication. Furthermore, this is not feasible in many countries.

Evidence about what to do once treatment targets are reached is also limited. While intravenous zoledronic acid may be the best option for many patients, the minimal effective dose and dosing interval to maintain BMD and fracture risk reduction, while avoiding rare complications, such as atypical femur fracture, is likely to vary among individuals and by prior osteoporosis treatment administered. In addition, some patients might not be candidates for bisphosphonates; these patients have very limited BMD maintenance options. Here, bisphosphonates are considered as a class with few distinctions among the agents, although most of the data available utilize alendronate or zoledronic acid.

Finally, including patients in decision-making regarding treatment targets, initial treatment selection and maintenance of treatment targets will likely add to the success of the goal directed therapy approach.^{69,70}

Conclusion

The ASBMR/BHOF Task Force on Goal-Directed Osteoporosis Treatment suggests that rapid fracture risk reduction should be the predominant treatment goal for patients at very high and imminent risk of fracture. The Task Force also recommends considering using BMD goals to guide clinical decisions for initiating and continuing treatment. Specific BMD targets and treatment decisions should be tailored to each patient's clinical risk profile and BMD. Considerations presented here may be modified in the future as more evidence becomes available.

Author contributions

Felicia Cosman (Conceptualization, Data curation, Project administration, Writing-original draft, Writing-review & editing), E. Michael Lewiecki (Conceptualization, Supervision, Writing-original draft, Writing-review & editing), Richard Eastell (Conceptualization, Writing-original draft, Writing-review & editing), Peter Ebeling (Conceptualization, Writing-original draft, Writing-review & editing), Suzanne Jan de Beur (Conceptualization, Writing-original draft, Writing-review & editing), Bente Langdahl (Conceptualization, Writing-original draft, Writing-review & editing), Yumie Rhee (Conceptualization, Writing-original draft, Writing-review & editing), Ghada El-Hajj Fuleihan (Writing-review & editing), Douglas Kiel (Writing-review & editing), John Schousboe, (Writing-review & editing), Joao Borges (Writing-review & editing), Angela Cheung (Writing-review & editing), Adolfo Diez-Perez (Writing-review & editing), Peyman Hadji (Writing-review & editing), Sakae Tanaka (Writing-review & editing), Friederike Thomasius (Writing-review & editing), Weibo Xia (Writing-review & editing), and Steven R. Cummings (Conceptualization, Supervision, Writing-original draft, Writing-review & editing).

Funding

The authors received no funding for this project.

Conflicts of interest

F.C.: Amgen: investigator, consultant, advisor, speaker; Radius Health: investigator, consultant, speaker; Enterabio, Biocon, CuraTeQ, Pfizer/ Myovant/Sumitomo, Theramex, UCB: consultant.

E.M.L.: Amgen: investigator, consultant, speaker; Radius: investigator, consultant; Ultragenyx: investigator; Kyowa Kirin: consultant, speaker; Angitia and Ascendis: consultant.

R.E.: AstraZeneca, Immunodiagnostic Systems, Sandoz, Samsung, CL Bio, CuraTeQ, Biocon, Takeda, UCB: consultant; Pharmacosmos, Alexion, UCB and Amgen: speaker; Alexion and Osteolabs: grants.

P.R.E.: Amgen: grant funds, speaker; Alexion: grant funds, speaker, consultant; Sanofi: investigator; Kyowa Kirin: speaker.

S.J.d.B.: Ultragenyx: investigator, consultant; Kyowa Kirin: investigator, consultant; Ascendis: consultant; Inozyme: consultant; BridgeBio: investigator; Mereo: Investigator. B.L.: Amgen: investigator, advisor, speaker; UCB: investigator, advisor, speaker; Gedeon-Richter: advisor, speaker; Astellas; speaker; Astra-Zeneca: speaker.

Y.R.: Amgen: investigator, speaker; Alexion: grant funds; Sanofi: investigator; Kyowa Kirin: investigator, speaker; Daewoong: investigator; Dongkook: investigator; Donga ST: investigator.

D.P.K.: Amgen: grant funds, Radius Health: consultant and grant funds, Agnovos: Data and Safety Monitoring Board.

G.E.H.F.: None.

J.T.S.: None.

J.L.C.B.: None.

A.C.: Amge n: investigator, consultant, speaker; Incyte: investigator; Ipsen: investigator, consultant, speaker; Sanofi: investigator.

A.D.P.: Theramex: speaker and advisor.

P.H.: Amgen, Besins, Exeltis, Gedeon Richter, Hexal, Sandoz, Stada, Theramex and UCB: speaker and consultant.

S.T.: Amgen, Asahi Kasei Pharma and Daiichi-Sankyo Co: speaker. F.T.: Amgen, Fresenius, Gedeon-Richter, Stada, Theramex, UCB: Advisor and speaker; Abbvie, Alexion: speaker.

W.X.: None.

S.R.C.: Amgen: speaker.

References

- Cummings S, Cosman F, Lewiecki E, et al. Goal-directed treatment for osteoporosis: a progress report from the ASBMR-NOF working group on goal-directed treatment for osteoporosis. J Bone Miner Res. 2017;32(1):3–10. https://doi.org/10.1002/jbmr.3039
- Camacho PM, Petak SM, Binkley N, et al. American Association of Clinical Endocrinologists/American College of Endocrinology clinical practice guidelines for the diagnosis and treatment of postmenopausal osteoporosis—2020 update. *Endocr Pract.* 2020;26(Suppl 1):1–46. https://doi.org/10.4158/GL-2020-0524 SUPPL
- Shoback D, Rosen CJ, Black DM, Cheung AM, Murad MH, Eastell R. Pharmacological management of osteoporosis in postmenopausal women: an Endocrine Society Guideline Update. J Clin Endo Metab. 2020;105(3):587–594. https://doi.org/10.1210/cline m/dgaa048
- 4. Kanis JA, Cooper C, Rizzoli R, Reginster JY, Scientific Advisory Board of the European Society for Clinical and Economic Aspects of Osteoporosis and Osteoarthritis (ESCEO) and the Committees of Scientific Advisors and National Societies of the International Osteoporosis Foundation (IOF). Executive summary of European guidance for the diagnosis and management of osteoporosis in postmenopausal women. *Aging Clin Exp Res.* 2019;31(1):15–17. https://doi.org/10.1007/s40520-018-1109-4
- Management of Osteoporosis in Postmenopausal Women: The 2021 Position Statement of the North American Menopause Society Editorial Panel. Management of osteoporosis in postmenopausal women: the 2021 position statement of the North American Menopause Society. *Menopause*. 2021;28(9):973–997. https://doi.org/10.1097/GME.00000000001831
- LeBoff M, Greenspan S, Insogna K, et al. The clinician's guide to prevention and treatment of osteoporosis. Osteoporos Int. 2022;33(10):2049–2102. https://doi.org/10.1007/s00198-021-05900-y
- Gregson CL, Armstrong DJ, Bowden J, et al. UK clinical guideline for the prevention and treatment of osteoporosis. *Arch Osteoporo*. 2022;17(1):58. https://doi.org/10.1007/s11657-022-01061-5
- Balasubramanian A, Zhang J, Chen L, et al. Risk of subsequent fracture after prior fracture among older women. Osteoporos Int. 2019;30(1):79–92. https://doi.org/10.1007/s00198-018-4732-1
- Schnell AD, Curtis JR, Saag KG. Importance of recent fracture as predictor of imminent fracture risk. *Curr Osteoporos Rep.* 2018;16(6):738–745.https://doi.org/10.1007/s11914-018-0487-z
- Banefelt J, Åkesson KE, Spångéus A, et al. Risk of imminent fracture following a previous fracture in a Swedish database study.

Osteoporos Int. 2019;30(3):601-609. https://doi.org/10.1007/s00198-019-04852-8

- 11. Johansson H, Siggeirsdottir K, Harvey NC, et al. Imminent risk of fracture after fracture. Osteoporos Int. 2017;28(3):775–780. https://doi.org/10.1007/s00198-016-3868-0
- Yusuf AA, Chandler HY, Crittenden DB, Barron RL. Predictors of imminent risk of fracture in Medicare-enrolled men and women. Arch Osteoporos. 2020;15(1):120. doi.org/10.1007/s11657-020-00784-7
- Adachi JD, Berger C, Barron R, et al. Predictors of imminent non-vertebral fracture in elderly women with osteoporosis, low bone mass, or a history of fracture, based on data from the population-based Canadian multicentre osteoporosis study (CaMos). *Arch Osteoporos.* 2019;14(1):53. https://doi.org/10.1007/s11657-019-0598-x
- Barron RL, Oster G, Grauer A, Crittenden DB, Weycker D. Determinants of imminent fracture risk in postmenopausal women with osteoporosis. *Osteoporos Int.* 2020;31(11):2103–2111. https://doi.org/10.1007/s00198-020-05294-3
- Lindsay R, Silverman SL, et al. Risk of new vertebral fracture in the year following a fracture. JAMA. 2001;285(3):320–323. https:// doi.org/10.1001/jama.285.3.320
- Johnell O, Kanis JA, Odén A, et al. Fracture risk following an osteoporotic fracture. Osteoporos Int. 2004;15(3):175–179. https://doi.org/10.1007/s00198-003-1514-0
- Delmas PD, Genant HK, Crans GG, et al. Severity of prevalent vertebral fractures and the risk of subsequent vertebral and nonvertebral fractures: results from the MORE trial. *Bone*. 2003;33(4): 522–532. https://doi.org/10.1016/S8756-3282(03)00241-2
- Gehlbach S, Saag KG, Adachi JD, et al. Previous fractures at multiple sites increase the risk for subsequent fractures: the global longitudinal study of osteoporosis in women. J Bone Miner Res. 2012;27(3):645–653. https://doi.org/10.1002/jbmr.1476
- Soreskog E, Strom O, Spangeus A, et al. Risk of major osteoporotic fracture after first, second and third in Swedish women aged 50 years and older. *Bone*. 2020;134:115286. https://doi.org/10.1016/ j.bone.2020.115286
- Beaudoin C, Jean S, Moore L, et al. Number, location, and time since prior fracture as predictors of future fracture in the elderly from the general population. J Bone Miner Res. 2018;33(11): 1956–1966. https://doi.org/10.1002/jbmr.3526
- Kendler DL, Marin F, Zerbini CAF, et al. Effects of teriparatide and risedronate on new fractures in post-menopausal women with severe osteoporosis (VERO): a multicentre, double-blind,double dummy, randomised controlled trial. *Lancet*. 2018;391(10117): 230–240. https://doi.org/10.1016/S0140-6736(17)32137-2
- 22. Saag KG, Petersen J, Brandi ML, et al. Romosozumab or alendronate for fracture prevention in women with osteoporosis. N Engl J Med. 2017;377(15):1417–1427. https://doi.org/10.1056/ NEJMoa1708322
- Leder BZ, Mitlak B, Hu MY, Hattersley G, Bockman RS. Effect of abaloparatide versus alendronate on fracture risk reduction in postmenopausal women with osteoporosis. J Clin Endo Metab. 2020;105(3):938–943. https://doi.org/10.1210/clinem/ dgz162
- Hadji P, Zanchetta JR, Russo L, et al. The effect of teriparatide compared with risedronate on reduction of back pain in postmenopausal women with osteoporotic vertebral fractures. *Osteoporos Int*. 2012;23(8):2141–2150. https://doi.org/10.1007/ s00198-011-1856-y
- Saag KG, Shane E, Boonen S, et al. Teriparatide or alendronate in glucocorticoid-induced osteoporosis. N Engl J Med. 2007;357(20): 2028–2039. https://doi.org/10.1056/NEJMoa071408
- Cosman F, Oates M, Betah D, et al. Romosozumab followed by Denosumab versus Denosumab only: a post-hoc analysis of FRAME and FRAME extension. J Bone Mineral Res. 2024:zjae116. https://doi.org/10.1093/jbmr/zjae116
- Bouxsein ML, Eastell R, Lui L-Y, et al. Change in Bone density and reduction in fracture risk: a meta-regression of published trials.

J Bone Miner Res. 2019;34(4):632-642. https://doi.org/10.1002/jbmr.3641

- Black DM, Bauer DC, Vittinghoff E, et al. Treatment-related changes in bone mineral density as a surrogate biomarker for fracture risk reduction: meta-regression analyses of individual patient data from multiple randomised controlled trials. *Lancet Diabetes Endocrinol*. 2020;8(8):672–682. https://doi.org/10.1016/ S2213-8587(20)30159-5
- Eastell R, Vittinghoff E, Lui L-Y, et al. Validation of the surrogate threshold effect for change in bone mineral density as a surrogate endpoint for fracture outcomes: the FNIH-ASBMR SABRE project. *J Bone Miner Res.* 2022;37(1):29–35. https://doi.org/10.1002/ jbmr.4433
- Black DM, Schwartz AV, Ensrud KE, et al. Effects of continuing or stopping alendronate after 5 years of treatment: the fracture intervention trial long-term extension (FLEX): a randomized trial. JAMA. 2006;296(24):2927–2938. https://doi.org/10.1001/ja ma.296.24.2927
- 31. Cosman F, Cauley JA, Eastell R, et al. Reassessment of fracture risk in women after 3 years of treatment with zoledronic acid: when is it reasonable to discontinue treatment? J Clin Endo Metab. 2014;99(12):4546–4554. https://doi.org/10.1210/jc.2014-1971
- 32. Ferrari S, Libanati C, Lin CJF, et al. Relationship between bone mineral density T-score and nonvertebral fracture risk over 10 years of denosumab treatment. J Bone Miner Res. 2019;34(6): 1033–1040. https://doi.org/10.1002/jbmr.3722
- 33. Cosman F, Lewiecki EM, Ebeling P, et al. T-score as an indicator of fracture risk on therapy: evidence from romosozumab versus alendronate treatment in the ARCH trial. J Bone Miner Res. 2020;35(7):1333–1342. https://doi.org/10.1002/jbmr.3996
- Cosman F, Nieves JW, Dempster DW. Treatment sequence matters: anabolic and antiresorptive therapy for osteoporosis. J Bone Miner Res. 2017;32(2):198–202. https://doi.org/10.1002/jbmr.3051
- 35. Langdahl BL, Libanati C, Crittenden DB, et al. Romosozumab (sclerostin monoclonal antibody) versus teriparatide in postmenopausal women with osteoporosis transitioning from oral bisphosphonate therapy: a randomised, open-label, phase 3 trial. *Lancet*. 2017;390(10102):1585–1594. https://doi.org/10.1016/ S0140-6736(17)31613-6
- Leder BZ, Tsai JN, Uihlein AV, et al. Denosumab and teriparatide transitions in postmenopausal osteoporosis (the DATAswitch study): extension of a randomised controlled trial. *Lancet*. 2015;386(9999):1147–1155. https://doi.org/10.1016/S0140-6736 (15)61120-5
- Cosman F, Kendler DL, Langdahl BL, et al. Romosozumab and antiresorptive treatment: the importance of treatment sequence. Osteoporos Int. 2022;33(6):1243–1256. https://doi.org/10.1007/ s00198-021-06174-0
- Cosman F, Langdahl B, Leder BZ. Treatment sequence for osteoporosis. *Endocr Pract*. 2024;30(5):490–496. https://doi.o rg/10.1016/j.eprac.2024.01.014
- Cosman F, Libanati C, Deignan C, et al. Romosozumab followed by antiresorptive treatment increases the probability of achieving bone mineral density treatment goals. J Bone Miner Res Plus. 2021;5(11):e10546. https://doi.org/10.1002/jbm4.10546
- Cosman F, Mitlak B, Wang Y, et al. Probability of achieving BMD treatment targets with abaloparatide and teriparatide over 18 months and sequential abaloparatide vs. placebo followed by alendronate over 3.5 years. J Bone Miner Res. 2022;37(Suppl 1). Accessed August 15, 2024. Available at https://www.asbmr.o rg/meetings/annualmeeting/AbstractDetail?aid=a22184c5-03 f3-4740-b021-46b8e35e5d85.
- 41. Cosman F, Huang S, Chines A, Cummings S. Probability of achieving bone mineral density treatment goals with denosumab treatment in women with osteoporosis. J Bone Miner Res. 2022;37(Suppl 1). Accessed August 15, 2024. Available at https://www.asbmr.org/meetings/annualmeeting/AbstractDetail? aid=6540c458-900f-4bea-8f3c-0d8c12208ec2.

- 42. https://iscd.org/wp-content/uploads/2024/03/2023-ISCD-Adult-Positions.pdf
- 43. Ferrari S, Lippuner K, Olivier L, Christian M. 2020 recommendations for osteoporosis treatment according to fracture risk from the Swiss association against osteoporosis (SVGO). *Swiss Med Wkly*. 2020;150(3940):w20352. https://doi.org/10.4414/ smw.2020.20352
- 44. Kanis JA, Oden A, Johnell O, et al. The use of clinical risk factors enhances the performance of BMD in the prediction of hip and osteoporotic fractures in men and women. *Osteoporos Int.* 2007;18(8):1033–1046. https://doi.org/10.1007/ s00198-007-0343-y
- 45. Black DM, Cauley JA, Wagman R, et al. The ability of a single BMD and fracture history assessment to predict fracture over 25 years in postmenopausal women: the study of osteoporotic fractures. *J Bone Miner Res.* 2018;33(3):389–395. https://doi.org/10.1002/ jbmr.3194
- El-Hajj Fuleihan G, Chaktoura M, Cauley JA, Chamoun N. Worldwide fracture prediction. J Clin Densitom. 2017;20(3):397–424. https://doi.org/10.1016/j.jocd.2017.06.008
- Bartosch P, Malmgren L, Kristensson J, McGuigan FE, Akesson KE. In community-dwelling women frailty is associated with imminent risk of osteoporotic fractures. Osteoporos Int. 2021;32(9): 1735–1744. https://doi.org/10.1007/s00198-021-05886-7
- Iconaru L, Charles A, Baleanu F, et al. Prediction of an imminent fracture after an index fracture models derived from the Frisbee cohort. J Bone Miner Res. 2022;37(1):59–67. https://doi.org/10.1002/jbmr.4432
- 49. Kim KM, Lui L-Y, Cummings SR. Recent fall and high imminent risk of fracture in older men and women. *Age Ageing*. 2022;51(6):afac141. https://doi.org/10.1093/ageing/afac141
- McClung MR, Bolognese MA, Brown JP, et al. Skeletal responses to romosozumab after 12 months of denosumab. J Bone Miner Res Plus. 2021;5(7):e10512. https://doi.org/10.1002/jbm4.10512
- Schuit SCE, van der Klift M, Weel AEAM, et al. Fracture incidence and association with bone mineral density in elderly men and women: the Rotterdam study. *Bone*. 2004;34(1):195–202. https:// doi.org/10.1016/j.bone.2003.10.001
- Wainwright SA, Marshall LM, Ensrud KE, et al. Hip fracture in women without osteoporosis. J Clin Endocrinol Metab. 2005;90(5):2787–2793. https://doi.org/10.1210/jc.2004-1568
- Binkley N, Morin SN, Martineau P, Lix LM, Hans D, Leslie WD. Frequency of normal bone measurement in postmenopausal women with fracture: a registry-based cohort study. *Osteoporos Int*. 2020;31(12):2337–2344. https://doi.org/10.1007/ s00198-020-05576-w
- 54. Schini M, Vilaca T, Lui L-Y, et al. Pre-treatment bone mineral density (BMD) and the benefit of pharmacologic treatment on fracture risk and BMD change: analysis from the FNIH-ASBMR SABRE project. J Bone Miner Res. 2024;39(7):867–876. https:// doi.org/10.1093/jbmr/zjae068
- Black DM, Thompson DE, Bauer DC, et al. Fracture risk reduction with alendronate in women with osteoporosis: the fracture intervention trial. J Clin Endo Metab. 2000;85(11):4118–4124
- Black DM, Delmas PD, Eastell R, et al. Once-yearly zoledronic acid 5 mg for treatment of postmenopausal osteoporosis. *NEJM*. 2007;356(18):1809–1822
- Cummings SR, San Martin J, McClung MR, et al. Denosumab for prevention of fractures in postmenopausal women with osteoporosis. N Engl J Med. 2009;361(8):756–765. https://doi.org/10.1056/ NEJMoa0809493
- Neer RM, Arnaud CD, Zanchetta JR, et al. Effect of parathyroid hormone (1-34) on fractures and bone mineral density in postmenopausal women with osteoporosis. N Engl J Med. 2001;344(19):1434–1441. https://doi.org/10.1056/NEJM 200105103441904
- 59. Miller PD, Hattersley G, Riis BJ, et al. Effect of Abaloparatide versus placebo on new vertebral fractures in postmenopausal women

with osteoporosis: a randomized clinical trial. *JAMA*. 2016;316(7): 722–733. https://doi.org/10.1001/jama.2016.11136

- Bone HG, Cosman F, Miller PD, et al. ACTIVExtend: 24 months of alendronate after 18 months of Abaloparatide or placebo for postmenopausal osteoporosis. J Clin Endo Metab. 2018;103(8): 2949–2957. https://doi.org/10.1210/jc.2018-00163
- Cosman F, Crittenden DB, Grauer A, et al. Romosozumab treatment in postmenopausal osteoporosis. N Engl J Med. 2017;376(4): 396–397. https://doi.org/10.1056/NEJMc1615367
- Lewiecki EM, Dinavahi RV, Lazaretti-Castro M, et al. One year of Romosozumab followed by two years of Denosumab maintains fracture risk reductions: results of the FRAME extension study. *J Bone Miner Res.* 2019;34(3):419–428. https://doi.org/10.1002/ jbmr.3622
- 63. Kendler DL, Chines A, Brandi ML, et al. The risk of subsequent osteoporotic fractures is decreased in subjects experiencing fracture while on denosumab: results from the FREEDOM and FREEDOM extension studies. Osteoporos Int. 2019;30(1):71–78. https://doi.o rg/10.1007/s00198-018-4687-2
- 64. Tsourdi E, Langdahl B, Cohen-Solal M, et al. Discontinuation of Denosumab therapy for osteoporosis: a systematic review and position statement by ECTS. *Bone*. 2017;105:11–17. https://doi.org/10.1016/j.bone.2017.08.003
- 65. Cosman F, Huang S, McDermott M, Cummings SR. Multiple vertebral fractures after Denosumab discontinuation: FREEDOM

and FREEDOM extension trials additional post hoc analyses. J Bone Miner Res. 2022;37(11):2112–2120. https://doi.org/10.1002/jbmr.4705

- 66. Ensrud KE, Barrett-Connor EL, Schwartz A, et al. Randomized trial of effect of alendronate continuation versus discontinuation in women with low BMD: results from the fracture intervention trial long-term extension. J Bone Miner Res. 2004;19(8):1259–1269. https://doi.org/10.1359/JBMR.040326
- 67. Black DM, Reid IR, Cauley JA, et al. The effect of 6 versus 9 years of zoledronic acid treatment in osteoporosis: a randomized second extension to the HORIZON-pivotal fracture trial (PFT). J Bone Miner Res. 2015;30(5):934–944. https://doi.org/10.1002/jbmr.2442
- Grey A, Bolland MJ, Horne A, et al. Five years of antiresorptive ativity after a single dose of zoledronate- results from a randomized double-blind placebo-controlled trial. *Bone*. 2012;50(6):1389–1393. https://doi.org/10.1016/j.bone.2012. 03.016
- Conley RB, Adib G, Adler RA, et al. Secondary fracture prevention: consensus clinical recommendations from a multi-stakeholder coalition. J Bone Miner Res. 2019;35(1):36–52. https://doi.org/10.1002/jbmr.3877
- El-Hajj FG. Building bridges to address the osteoporosis crisis. J Clin Endo Metab. 2020;105(4):1292–1293. https://doi.o rg/10.1210/clinem/dgz307